%~MoulinÃ© par MaN_auto v.0.34.0 (b272df1a) 2025-06-13 15:10:23
\documentclass[CRMATH,Unicode,XML,biblatex]{cedram}

\TopicFR{AlgÃ¨bre}
\TopicEN{Algebra}

\addbibresource{CRMATH_Guo_20230936.bib}

\usepackage[all]{xy}
\SelectTips{eu}{}

\newcommand{\ZZ}{\mathbb{Z}}

\DeclareMathOperator{\Kerop}{Ker}
\newcommand{\Ker}[1]{\Kerop #1}

\DeclareMathOperator{\Cokerop}{Coker}
\newcommand{\Coker}[1]{\Cokerop #1}

\DeclareMathOperator{\Hop}{H}
\newcommand{\HH}[2][]{\Hop^{#1}(#2)}

\DeclareMathOperator{\Homop}{Hom}
\newcommand{\Hom}[3][R]{\Homop_{#1}(#2,#3)}

\DeclareMathOperator{\Extop}{Ext}
\newcommand{\Ext}[4][R]{\Extop_{\smash{#1}}^{#2}(#3,#4)}
\newcommand{\Cext}[4][R]{\smash{\widehat{\Extop}}^{#2}_{\smash{#1}}(#3,#4)}
\newcommand{\Text}[4][R]{\smash{\widecheck{\Extop}}_{\smash{#1}}^{#2}(#3,#4)}

\DeclareMathOperator{\F}{F}
\newcommand{\TF}[1]{\smash{\widecheck{\F}}^{#1}}
\newcommand{\tF}[1]{\smash{\widecheck{\F(N)}}^{#1}}

\DeclareMathOperator{\G}{G}
\newcommand{\TG}[1]{\smash{\widehat{\G}}^{#1}}
\newcommand{\tG}[1]{\smash{\widehat{\G(M)}}^{#1}}

\newcommand{\suspc}{\mathsf{\Theta}_{\C}}
\newcommand{\suspy}{\mathsf{\Theta}_{\y}}

\DeclareMathOperator{\sa}{S}
\DeclareMathOperator{\T}{T}
\DeclareMathOperator{\U}{U}

\newcommand{\D}{\mathsf{W}}
\newcommand{\DR}{\D R}

\newcommand{\z}{\mathsf{Z}}
\newcommand{\C}{\mathsf{V}}
\newcommand{\x}{\mathsf{X}}
\newcommand{\y}{\mathsf{Y}}
\newcommand{\A}{\mathsf{A}}
\newcommand{\Ab}{\mathsf{Ab}}
\newcommand{\Inj}{\mathsf{Inj}}
\newcommand{\Prj}{\mathsf{Prj}}
\newcommand{\Mod}{\mathsf{Mod}}

\newcommand{\RMod}{R\text{-}\Mod}
\newcommand{\CHom}[3][R]{\smash{\overline{\Homop}}^{\C}_{#1}(#2,#3)}
\newcommand{\DHom}[3][R]{\underline{\Homop}^{\D}_{#1}(#2,#3)}
\newcommand{\cHom}[3][\A]{\smash{\overline{\Homop}}^{\C}_{#1}(#2,#3)}
\newcommand{\dHom}[3][\A]{\underline{\Homop}^{\D}_{#1}(#2,#3)}

\DeclareMathOperator{\idop}{id}
\newcommand{\id}[2][R]{\idop_{#1}#2}

\DeclareMathOperator{\SoneTop}{S_1T}
\DeclareMathOperator{\Image}{Im}
\DeclareMathOperator{\Cop}{C}
\DeclareMathOperator{\Zop}{Z}
\DeclareMathOperator{\pdop}{pd}

\DeclareMathOperator{\Dpdop}{\D-\pdop}
\DeclareMathOperator{\Cidop}{\C-\idop}
\DeclareMathOperator{\yidop}{\y-\idop}
\DeclareMathOperator{\xpdop}{\x-\pdop}
\DeclareMathOperator{\colim}{colim}

%%% \widecheck (pour aller avec \widehat)
\DeclareFontFamily{U}{mathx}{}
\DeclareFontShape{U}{mathx}{m}{n}{<-> mathx10}{}
\DeclareSymbolFont{mathx}{U}{mathx}{m}{n}
\DeclareMathAccent{\widehat}{0}{mathx}{"70}
\DeclareMathAccent{\widecheck}{0}{mathx}{"71}

%%%---------------------------------------------------------------------------

%%% Change counter appearance
\newcommand*{\relabel}{\renewcommand{\labelenumi}{(\theenumi)}}
\newcommand*{\romanenumi}{\renewcommand*{\theenumi}{\roman{enumi}}\relabel}
\newcommand*{\alphenumi}{\renewcommand*{\theenumi}{\alph{enumi}}\relabel}

%%%---------------------------------------------------------------------------

%%% ''Such that'' macro
\newcommand*{\st}[1][]{\:#1\vert\:}

%%%---------------------------------------------------------------------------

%%% ARROWS

%%% \to
\renewcommand*{\to}{\mathchoice{\longrightarrow}{\rightarrow}{\rightarrow}{\rightarrow}}
%% Remplacer \xrightarrow{#1} par \overset{#1}{\longrightarrow}. 

%%% \mapsto
%\let\oldmapsto\mapsto
%\renewcommand*{\mapsto}{\mathchoice{\longmapsto}{\oldmapsto}{\oldmapsto}{\oldmapsto}}

%%% \longhookrightarrow
%\newcommand{\longhookrightarrow}{\!\lhook\joinrel\longrightarrow}

%%% \longrightrightarrows
%\newcommand{\longrightrightarrows}{\overset{\raisebox{-.17ex}{$\displaystyle{\longrightarrow}$}}{\smash{\raisebox{-.3ex}{$\longrightarrow$}}}}

%%%% \longrightharpoonup
%\newcommand{\longrightharpoonup}{\relbar\joinrel\rightharpoonup}

%%% Convergence
%\newcommand\dconverge[2][p \to + \infty]{\xrightarrow[#1]{#2}}
%\newcommand\tconverge[1][p \to + \infty]{\underset{#1}{\longrightarrow}}

%%% Isomorphism
\newcommand{\qra}{\;\overset{\!\!\raisebox{.6ex}{$\scriptstyle{\simeq}$}}{\smash{\longrightarrow}}\:}

\newcommand{\xra}[1]{\;\overset{\!\!\raisebox{.8ex}{$\scriptstyle{#1}$}}{\smash{\longrightarrow}}\:}

%\newcommand{\xralow}[1]{\;\overset{\!\!\raisebox{.5ex}{$\scriptstyle{#1}$}}{\smash{\longrightarrow}}\:}

%\newcommand{\xra}[2][]{\xrightarrow[#1]{\;#2\;}}
%\newcommand{\qra}{\xra{\smash{\simeq}}}

%%%---------------------------------------------------------------------------

%%% DELIMITERS

\usepackage{mathtools}
\DeclarePairedDelimiter{\parens}{\lparen}{\rparen}
\DeclarePairedDelimiter{\braces}{\{}{\}}
%\DeclarePairedDelimiter{\abs}{\lvert}{\rvert}
%\DeclarePairedDelimiter{\norm}{\lVert}{\rVert}
\DeclarePairedDelimiter{\bracks}{[}{]}
%\DeclarePairedDelimiter{\bbracks}{\llbracket}{\rrbracket}
%\DeclarePairedDelimiter{\angles}{\langle}{\rangle}
%\DeclarePairedDelimiter{\aangles}{\langle\!\langle}{\rangle\!\rangle}
%\DeclarePairedDelimiter{\pparens}{(\!(}{)\!)}

%%%---------------------------------------------------------------------------

\graphicspath{{./figures/}}

\newcommand*{\mk}{\mkern -1mu}
\newcommand*{\Mk}{\mkern -2mu}
\newcommand*{\mK}{\mkern 1mu}
\newcommand*{\MK}{\mkern 2mu}

\hypersetup{urlcolor=purple, linkcolor=blue, citecolor=red}

\let\oldtilde\tilde
\renewcommand*{\tilde}[1]{\mathchoice{\widetilde{#1}}{\widetilde{#1}}{\oldtilde{#1}}{\oldtilde{#1}}}
\let\oldhat\hat
\renewcommand*{\hat}[1]{\mathchoice{\widehat{#1}}{\widehat{#1}}{\oldhat{#1}}{\oldhat{#1}}}

%%%---------------------------------------------------------------------------

%%% Changer la largeur des adresses

% Jouer avec \hsize :
\makeatletter
\def\@setafterauthor{%
  \vglue3mm%
%  \hspace*{0pt}%
\begingroup\hsize=11cm\advance\hsize\abstractmarginL\raggedright
\noindent
%\hspace*{\abstractmarginL}\begin{minipage}[t]{10cm}
   \leftskip\abstractmarginL
  \normalfont\Small
  \@afterauthor\par
\endgroup
\vskip2pt plus 3pt minus 1pt
}
\makeatother

%%%---------------------------------------------------------------------------

\title{Balancedness of complete cohomology with respect to subcategories}
\alttitle{Propri{\'e}t{\'e} d'{\'e}quilibre de la cohomologie compl{\`e}te par rapport aux sous-cat{\'e}gories}

\author{\firstname{Shoutao} \lastname{Guo}}
\address{Department of Mathematics, Gansu Center for Fundamental Research in Complex Systems Analysis and Control, Lanzhou Jiaotong University, Lanzhou 730070, China}
\email[S.~Guo]{guoshoutao9022@163.com}

\author{\firstname{Li} \lastname{Liang}\IsCorresp}
\address[1]{Department of Mathematics, Gansu Center for Fundamental Research in Complex Systems Analysis and Control, Lanzhou Jiaotong University, Lanzhou 730070, China}
\address{Gansu Provincial Research Center for Basic Disciplines of Mathematics and Statistics, Lanzhou 730070, China}
\email[L.~Liang]{lliangnju@gmail.com}
\urladdr{https://sites.google.com/site/lliangnju}

\author{\firstname{Xiaoyan} \lastname{Yang}}
\address{School of Science, Zhejiang University of Science and Technology, Hangzhou, 310012, China}
\email[X.~Yang]{yangxiaoyanzj@outlook.com}

\thanks{S.~Guo was partly supported by the Joint Innovation Fund Project of Lanzhou Jiaotong University and Beijing Jiaotong University (Grant No.~LH2024017); L.~Liang was partly supported by NSF of China (Grant No.~12271230) and the Foundation for Innovative Fundamental Research Group Project of Gansu Province (Grant Nos.~23JRRA684 and~25JRRA805); X.~Yang was partly supported by Conventional Projects for Graduate Education Reform in the Second Batch of the 14th Five Year Plan in Zhejiang Province (Grant No.~JGCG2024341)}
\CDRGrant[Joint Innovation Fund Project of Lanzhou Jiaotong University and Beijing Jiaotong University]{LH2024017}
\CDRGrant[NSF of China]{12271230}
\CDRGrant[Foundation for Innovative Fundamental Research Group Project of Gansu Province]{23JRRA684}
\CDRGrant[Foundation for Innovative Fundamental Research Group Project of Gansu Province]{25JRRA805}
\CDRGrant[Conventional Projects for Graduate Education Reform in the Second Batch of the 14th Five Year Plan in Zhejiang Province]{JGCG2024341}

\keywords{\kwd{Satellite} \kwd{balancedness} \kwd{complete cohomology} \kwd{Tate cohomology}}
\altkeywords{\kwd{Satellite} \kwd{{\'e}quilibre} \kwd{cohomologie compl{\`e}te} \kwd{cohomologie de Tate}}
\subjclass{18G25, 16E30, 16E05}

\begin{abstract}
We study the balancedness of complete cohomology with respect to a special preenveloping/precovering subcategory. As an application, a balanced result for Tate cohomology is given.
\end{abstract}

\begin{altabstract}
Dans cet article, nous Ã©tudions la propriÃ©tÃ© d'Ã©quilibre de la cohomologie complÃ¨te par rapport Ã  une sous-catÃ©gorie spÃ©ciale prÃ©-enveloppante/prÃ©-couvrante. Comme application, nous dÃ©montrons une propriÃ©tÃ© d'Ã©quilibre pour la cohomologie de Tate.
\end{altabstract}

\begin{document}

\maketitle

\section*{Introduction}

Tate cohomology can be traced back to the work of Tate in 1952 (see~\cite{JT}), which was initially defined for representations of finite groups. Avramov and Martsinkovsky~\cite{AM} extended the definition so that it can work well for finitely generated modules of finite Gorenstein dimension over a noetherian ring. Sather-Wagstaff, Sharif and White~\cite{SSW-10a} further investigated Tate cohomology of objects admitting a Tate resolution in an abelian category.

As a broad generalization of Tate cohomology to the realm of infinite group algebras or even associative rings, complete cohomology was introduced by Vogel and Goichot~\cite{GF92}, Mislin~\cite{GM94} and Benson and Carlson~\cite{BC92} independently, and was further treated by Avramov and Veliche~\cite{AL07} and Nucinkis~\cite{BN98}. Complete cohomology is an important subject in representation theory, commutative algebra and homological algebra. We mention that the $0$-th complete cohomology group $\Cext[R]{0}{M}{N}$ for $R$-modules $M$ and $N$ is actually the group of homomorphisms of objects in a stabilization of the category of $R$-modules, so Avramov and Veliche referred to complete cohomology as stable cohomology to emphasize this fact; see~\cite{AL07}. Recently, the first two authors of this paper introduced and studied the complete cohomology with respect to a special preenveloping/precovering subcategory via stable functors in~\cite{GL22}, where some properties including vanishing were given. This paper is a follow-up to~\cite{GL22}; we aim to investigate the balancedness of the relative complete cohomology.

We notice that the balancedness of the relative complete cohomology is given relying on the concept of balanced pairs introduced by Chen in~\cite{Chen}; the name ``balanced pair'' was settled to emphasize the capacity to induce balancedness in relative cohomology groups; see Remark~\ref{balance of relative cohomology}. Many examples of balanced pairs are from complete hereditary cotorsion triplets, as it was proved by Estrada, P\'{e}rez and Zhu~\cite[Proposition~4.2]{EPZ} that if $(\D,\z,\C)$ is a complete hereditary cotorsion triplet then $(\D,\C)$ is a balanced pair. In the following we give more details on the main results in the paper.

Throughout this paper, $R$ denotes an associative unital ring. By an $R$-module we mean a left $R$-module. The category of $R$-modules is denoted $\RMod$. We use the term ``subcategory'' to mean a full and additive subcategory that is closed under isomorphisms.

Let $\C$ be a special preenveloping subcategory of $\RMod$ and $\D$ a special precovering subcategory of $\RMod$, and let $M$ and $N$ be $R$-modules. We defined in~\cite[Definitions~3.4 and~3.7]{GL22} the complete cohomology groups $\Text[\C]{n}{M}{N}$ and $\Cext[\D]{n}{M}{N}$ as
\[
\Text[\C]{n}{M}{N}=\colim_{i}\cHom[R]{\suspc^{i}M}{\suspc^{i+n}N}
\]
and
\[
\Cext[\D]{n}{M}{N}=\colim_{i} \dHom[R]{\Omega^{\D}_{i+n}M}{\Omega^{\D}_{i}N}.
\]
In the paper, we study the relation between the group $\Text[\C]{n}{M}{N}$ and the group $\Cext[\D]{n}{M}{N}$, and prove the next result; see Theorems~\ref{balance} and~\ref{Tate result}.

\begin{enonce*}{Main Theorem}
Let $(\D,\C)$ be a balanced pair with $\D$ special precovering and $\C$ special preenveloping. Assume that $\D$ and $\C$ are closed under direct summands and ${\Ext{\geq1}{\D}{\D^{\bot}}}=0={\Ext{\geq1}{{^{\perp}\C}}{\C}}$. Then the following statements are equivalent.
\begin{enumerate}\romanenumi
\item All $R$-modules in $\C$ have finite $\D$-projective dimension and all $R$-modules in $\D$ have finite $\C$-injective dimension.
\item $\D$-$\pdop(\C)= \C$-$\idop(\D)<\infty$.
\item Each $R$-module has a Tate $\C$-coresolution.
\item Each $R$-module has a Tate $\D$-resolution.
\item For all $R$-modules $M$ and $N$ and each $n\in\ZZ$, there is a natural isomorphism
\[
\Text[\C]{n}{M}{N} \cong \Cext[\D]{n}{M}{N}.
\]
\item For each $R$-module $M$ in $\C$ or\/ $\D$, there is a natural isomorphism
\[
\Text[\C]{0}{M}{M} \cong \Cext[\D]{0}{M}{M}.
\]
\end{enumerate}
\end{enonce*}

Here, $\Dpdop{(\C)}=\sup\braces[\big]{\Dpdop_{R}{(C)} \st[\big] C \in \C}$ and $\Cidop{(\D)}=\sup\braces[\big]{{\Cidop}_{R}{(D)} \st[\big] D \in \D}$. The definitions of $\D$-projective dimension $\Dpdop_{R}$ and $\C$-injective dimension $\Cidop_{R}$ are given in Remark~\ref{def dim}.

The above theorem is proved relying on the completion of cohomology sequences studied in Section~\ref{completion} and some vanishing results given in~\cite{GL22}. As an application, we give a new balanced result for the Tate cohomology; it is proved that under the equivalent conditions in Main Theorem there is a natural isomorphism $\Cext[\DR]{n}{M}{N} \cong \Cext[R\C]{n}{M}{N}$ (see Theorem~\ref{Tate result}).

\section{Preliminaries}

In this section we mainly recall some necessary notions and definitions.

\begin{enonce*}[remark]{Special preenveloping/precovering subcategories}
Given a subcategory $\x$ of $\RMod$, we write
\[
{^{\bot}{\x}}=\braces[\big]{M \st[\big] \Ext[R]{1}{M}{X}=0 \ \text{\normalfont for all} \ X\in\x} \qquad \text{and} \qquad
{\x}^{\bot}=\braces[\big]{N \st[\big] \Ext[R]{1}{X}{N}=0 \ \text{\normalfont{for all}} \ X\in\x}.
\]
Here $\Ext[R]{1}{-}{-}$ is the $1$st right derived functor of $\Hom{-}{-}$. Following Enochs and Jenda~\cite{EJ20}, a \emph{special\/ $\x$-preenvelope} of an $R$-module $N$ is an exact sequence $0 \to N \to X \to C \to 0$ with $X\in\x$ and $C\in{^{\bot}{\x}}$. Dually, a \emph{special $\x$-precover} of an $R$-module $M$ is an exact sequence $0 \to K \to X' \to M \to 0$ with $K\in{\x}^{\bot}$ and $X'\in\x$. Recall that a subcategory $\x$ of $\RMod$ is \emph{special preenveloping} if each $R$-module has a special $\x$-preenvelope. Dually a subcategory $\x$ of $\RMod$ is called \emph{special precovering} if each $R$-module has a special $\x$-precover. It is clear that the subcategory $\Inj$ of injective $R$-modules is special preenveloping, and the subcategory $\Prj$ of projective $R$-modules is special precovering.
\end{enonce*}

\begin{enonce*}[remark]{Proper (co)resolutions}
Let $\y$ be a subcategory of $\RMod$. A \emph{proper $\y$-coresolution} of an $R$-module $N$ is a complex $I$ of $R$-modules in $\y$ such that $I^{-n}=0=\HH[n]{I}$ for all $n>0$ and $\HH[0]{I}\cong N$, and the associated exact sequence
\[
I^{+}\equiv 0 \to N \to I^{0} \to I^1 \to \dotsb
\]
is $\Hom{-}{\y}$-exact (that is, it remains exact after applying the functor $\Hom{-}{Y}$ to it for each $Y\in\y$), which is always denoted $N \qra I$.

Let $\x$ be a subcategory of $\RMod$. A \emph{proper $\x$-resolution} of an $R$-module $M$ is a complex $P$ of $R$-modules in $\x$ such that $P_{-n}=0=\HH[n]{P}$ for all $n>0$ and $\HH[0]{P}\cong M$, and the associated exact sequence
\[
P^{+}\equiv \dotsb \to P_{1} \to P_{0} \to M \to 0
\]
is $\Hom{\x}{-}$-exact, which is always denoted $P \qra M$.
\end{enonce*}

\begin{remark}[Dimensions and relative cohomology]\label{def dim}
The \emph{$\y$-injective dimension} of an $R$-module $N$ is the quantity
\[
\yidop_{R}(N)=\inf\braces[\Big]{\sup\braces[\big]{n\geq0 \st[\big] I_n\neq0} \st[\big] N\qra I \ \text{\normalfont is a proper $\y$-coresolution of $N$}}.
\]
When $\y$ is the subcategory of injectives, $\y$-$\idop_{R}(N)$ is the classical injective dimension.

Let $N$ be an $R$-module with $N\qra I$ a proper $\y$-coresolution. Then for each $R$-module $M$ and every $i\in\ZZ$, the $i$-th \emph{relative $\y$-cohomology} of $N$ with coefficients in $M$ is defined as
\[
\Ext[R\y]{i}{M}{N}=\Hop^i\parens[\big]{\Homop_R(M,I)}.
\]

Specially, if $\x$ is the subcategory of injectives, then $\Ext[R\y]{i}{M}{N}$ is actually the cohomology group $\Ext[R]{i}{M}{N}$.

Dually, one has the definition of \emph{$\x$-projective dimension}, $\xpdop_{R}(M)$, of an $R$-module $M$. Also, for $R$-modules $M$ and $N$ with $P\qra M$ a proper $\x$-resolution, the $i$-th \emph{relative $\x$-cohomology} of $M$ with coefficients in $N$ is defined as
\[
\Ext[{\x} R]{i}{M}{N}=\Hop^i\parens[\big]{\Homop_R(P,N)}.
\]
\end{remark}

The next result can be found in~\cite[Lemma~2.4]{Chen}.

\begin{lemm}\label{Chen}
Let $M$ be an $R$-module. Then for each $n\geq0$, the following statements are equivalent.
\begin{enumerate}\romanenumi
\item $\x$-$\pdop_{R}(M)\leq n$.
\item $\Ext[{\x} R]{i}{M}{-}=0$ for all $i>n$.
\item For each proper $\x$-resolution $P\qra M$, $\Coker{(P_{n+1}\to P_n)}$ is in $\x$.
\end{enumerate}
\end{lemm}

Dually, one has the following result.

\begin{lemm}\label{Chen-dual}
Let $N$ be an $R$-module. Then for each $n\geq0$, the following statements are equivalent.
\begin{enumerate}\romanenumi
\item $\y$-$\idop_{R}(N)\leq n$.
\item $\Ext[R\y]{i}{-}{N}=0$ for all $i>n$.
\item For each proper $\y$-coresolution $N\qra I$, $\Ker{(I^n\to I^{n+1})}$ is in $\y$.
\end{enumerate}
\end{lemm}

\begin{enonce*}[remark]{(Co)Syzygies}
A proper $\y$-coresolution $N\qra I$ of an $R$-module $N$ is called \emph{special} if each $\Zop^i(I)=\Ker{(I^i\to I^{i+1})}$ is in ${^{\perp}\y}$ for $i\geq1$. We let $\suspy^{i}N$ denote the kernel $\Zop^i(I)$ for some special proper $\y$-coresolution $N\qra I$; it is always called the $i$-th \emph{$\y$-cosyzygy} of $N$.

Dually, a proper $\x$-resolution $P\qra M$ of an $R$-module $M$ is called \emph{special} if each $\Cop_i(P)=\Coker{(P_{i+1}\to P_i)}$ is in $\x^{\perp}$ for $i\geq1$. We let $\Omega^{\x}_{i}M$ denote the cokernel $\Cop_i(P)$ for some special proper $\x$-resolution $P\qra M$; it is always called the $i$-th \emph{$\x$-syzygy} of $M$.
\end{enonce*}

\begin{rema}\label{V and W}
We always set $\suspy N = \suspy^1 N$ and $\Omega^{\x}M=\Omega^{\x}_{1}M$. The fact that $\suspy^{i}N$ is in $^{\perp}\y$ and $\Omega^{\x}_{i}M$ is in $\x^{\perp}$ for each $i\geq1$ is used frequently in the paper.
\end{rema}

\begin{enonce*}[remark]{Setup}
Throughout this paper, the symbol $\D$ denotes a special precovering subcategory of $\RMod$, and the symbol $\C$ denotes a special preenveloping subcategory of $\RMod$. In this case, every $R$-module has a special proper $\D$-resolution and a special proper $\C$-coresolution.
\end{enonce*}

\section{Satellites and completion of cohomology sequences}\label{completion}

In this section, we give a description of the completion of cohomology sequences via satellites of functors. Throughout the section, $\F$ (resp., $\G$) denotes a contravariant (resp., covariant) additive functor from $\RMod$ to the abelian group category $\Ab$.

\begin{enonce*}[remark]{Left satellites}
For each $R$-module $M$, there is a special $\C$-preenvelope
\[
0\to M\to I\xra{\pi} \suspc M\to 0
\]
of $M$ with $I\in\C$ and $\suspc M\in{^{\perp}\C}$. Following Cartan and Eilenberg~\cite{CE56}, the first \emph{left satellite} of $\F$ with respect to $\C$, denoted ${\sa^{-1}_{\C}}\F$, is defined as ${\sa^{-1}_{\C}}\F(M)=\Ker \F(\pi)$. Then ${\sa^{-1}_{\C}}\F$ is a contravariant additive functor from $\RMod$ to $\Ab$, and it is independent of the choices of special $\C$-preenvelopes.

We set ${\sa^{-n}_{\C}}\F=\sa^{-1}_{\C}({\sa^{-n+1}_{\C}}\F)$ for each $n>0$, and set ${\sa^{0}_{\C}}\F=\F$.

Dually, for each $R$-module $M$, there is a special $\D$-precover
\[
0\to \Omega^{\D}M \xra{\epsilon} P\to M \to 0
\]
of $M$ with $P\in\D$ and $\Omega^{\D}M\in\D^{\bot}$. The first \emph{left satellite} of $\G$ with respect to $\D$, denoted ${\sa^{-1}_{\D}}\G$, is defined as ${\sa^{-1}_{\D}}\G(M)=\Ker \G(\epsilon)$. Then ${\sa^{-1}_{\D}}\G$ is a covariant additive functor from $\RMod$ to $\Ab$, and it is independent of the choices of special $\D$-precovers.

We set ${\sa^{-n}_{\D}}\G=\sa^{-1}_{\D}({\sa^{-n+1}_{\D}}\G)$ for each $n>0$, and set ${\sa^{0}_{\D}}\G=\G$.
\end{enonce*}

\begin{rema}\label{A2}
Since ${\sa^{-n}_{\C}}\F(I)=0$ for each $I\in\C$ and any $n>0$, there is a natural isomorphism
\[
{\sa^{-n}_{\C}}\F(M)\cong {\sa^{-n+k}_{\C}}\F(\suspc^{k}M)
\]
for $n>k\geq0$. Similarly, since ${\sa^{-n}_{\D}}\G(P)=0$ for each $P\in\D$ and all $n>0$, there is a natural isomorphism
\[
{\sa^{-n}_{\D}}\G(M)\cong {\sa^{-n+k}_{\D}}\G(\Omega^{\D}_{k}M)
\]
for $n>k\geq0$.
\end{rema}

The functor $\F$ is called \emph{half\/ $\C$-exact} if for each short exact sequence $0\to A'\to A\to A''\to 0$ of $R$-modules with $A''\in{^{\perp}\C}$, the sequence $\F(A'')\to \F(A)\to \F(A')$ is exact. The functor $\G$ is called \emph{half\/ $\D$-exact} if for each short exact sequence $0\to A'\to A\to A''\to 0$ of $R$-modules with $A'\in\D^{\bot}$, the sequence $\G(A')\to \G(A)\to \G(A'')$ is exact.

\begin{lemm}\label{half-exact}
The following statements hold.
\begin{enumerate}\alphenumi
\item \label{half-exact_a} If\/ $\F$ is half\/ $\C$-exact, then so is\/ $\sa_{\C}^{-i}\F(-)$ for each $i\geq1$.
\item \label{half-exact_b} If\/ $\G$ is half\/ $\D$-exact, then so is\/ $\sa_{\D}^{-i}\G(-)$ for each $i\geq1$.
\end{enumerate}
\end{lemm}

\begin{proof}
We only prove~\eqref{half-exact_a}; the statement~\eqref{half-exact_b} is proved similarly.

It is enough to prove that $\sa_{\C}^{-1}\F(-)$ is half $\C$-exact. Let
\[
0 \to M' \to M \to M'' \to 0
\]
be an exact sequence of $R$-modules with $M''\in{^{\perp}\C}$. Fix special $\C$-preenvelopes $0 \to M' \to I' \to \suspc M' \to 0$ and $0 \to M'' \to I'' \to \suspc M'' \to 0$. It follows from~\cite[Remark~8.2.2]{EJ20} that there is a commutative diagram with exact rows and columns:
\[
\xymatrix@C=20pt@R=15pt{ &0 \ar[d]_{} &0 \ar[d]_{} &0 \ar[d]_{} &&\\
0\ar[r]^{} &M'\ar[d]\ar[r]^{} & M\ar[d] \ar^{}[r] &M''\ar[d]\ar[r]^{ } &0 & \\
0\ar[r]^{} &I'\ar[d]\ar[r]^{} &I'\oplus I''\ar[d]\ar^{}[r] &I''\ar[d]\ar[r]^{ } &0 & \\
0\ar[r]^{}&\suspc M'\ar[r]_{}\ar[d]_{} &\suspc M \ar[r]\ar[d]_{} &\suspc M''\ar[r]^{}\ar[d]_{} &0\\
& 0 & 0 &0 & & }
\]
where the middle column is a special $\C$-preenvelope of $M$, as $\suspc M'$ and $\suspc M''$ are in $^{\perp}\C$ (see Remark~\ref{V and W}). Applying the functor $\F$ to the above diagram, one gets the next commutative diagram with columns exact:
\[
\xymatrix@C=20pt@R=15pt{ &0 \ar[d]_{} &0 \ar[d]_{} &0 \ar[d]_{} &&\\
&\sa_{\C}^{-1}\F(M'')\ar[d]\ar[r]^{} & \sa_{\C}^{-1}\F(M)\ar[d]\ar^{}[r]&\sa_{\C}^{-1}\F(M')\ar[d]&\\
&\F(\suspc M'')\ar[d]\ar[r]^{} &\F(\suspc M)\ar[d]\ar^{}[r] &\F(\suspc M')\ar[d] & \\
&\F(I'')\ar[r]_{} &\F(I'')\oplus\F(I') \ar[r] &\F(I').}
\]
The second non-zero row is exact as $\F$ is half $\C$-exact, and the third one is exact clearly. Thus by~\cite[III. Lemma~3.2]{CE56} the sequence $\sa_{\C}^{-1}\F(M'')\to \sa_{\C}^{-1}\F(M) \to \sa_{\C}^{-1}\F(M')$ is exact.
\end{proof}

The following result is a relative version of~\cite[Theorem~3.1]{CE56}. We mention that there is a mistake in the proof of~\cite[Theorem~3.1]{CE56}, which was discovered and fixed by Flanders in~\cite{HF64}. More precisely, an error occurred while proving the exactness of the sequence ${\SoneTop}(A'') \to \T(A') \to \T(A)$ in~\cite[Theorem~3.1]{CE56}. The authors claimed that the connecting homomorphism ${\SoneTop}(A'') \to \T(M)$ obtained by the Snake Lemma coincides with the inclusion map ${\SoneTop}(A'')\hookrightarrow \T(M)$. Flanders analysed the reason in~\cite{HF64} for the inconsistency of the above two homomorphisms; the main reason is that the homomorphisms $\gamma$ and $\mu$ in the commutative diagram in~\cite[p.~834]{HF64} are not the same. Furthermore, Flanders proved directly the exactness of the sequence ${\SoneTop}(A'') \to \T(A') \to \T(A)$ using the diagram chasing. For the convenience of the readers, we provide a detailed proof for the next result.

\begin{prop}\label{complex sequence}
The following statements hold.
\begin{enumerate}\alphenumi
\item \label{complex_sequence_a} Let\/ $\F$ be half\/ $\C$-exact. If\/ $0 \to M' \to M \to M'' \to 0$ is a short exact sequence of $R$-modules with $M''\in{^{\perp}\C}$, then there is an exact sequence
\[
\dotsb \to \sa_{\C}^{-1}\F(M) \to \sa_{\C}^{-1}\F(M') \to \F(M'') \to \F(M) \to \F(M').
\]
\item \label{complex_sequence_b} Let\/ $\G$ be half\/ $\D$-exact. If\/ $0 \to N' \to N \to N'' \to 0$ is a short exact sequence of $R$-modules with $N'\in \D^{\bot}$, then there is an exact sequence
\[
\dotsb \to \sa_{\D}^{-1}\G(N) \to \sa_{\D}^{-1}\G(N'') \to \G(N')\to \G(N) \to \G(N'').
\]
\end{enumerate}
\end{prop}

\begin{proof}
We only prove~\eqref{complex_sequence_a}; the statement~\eqref{complex_sequence_b} is proved similarly. The proof of the exactness of the sequence $\sa_{\C}^{-1}\F(M') \to \F(M'') \to \F(M)$ is similar to the one in~\cite{HF64}. We use a slightly different proof to prove the exactness of the sequence $\sa_{\C}^{-1}\F(M) \to \sa_{\C}^{-1}\F(M') \to \F(M'')$, which seems more direct to us.

Fix a special $\C$-preenvelope $0 \to M \to I \to \suspc M \to 0$, and consider the following pushout diagram:
\begin{equation}\label{diag01}
\begin{aligned}
\xymatrix@C=12pt@R=15pt{ &&0\ar[d]&0\ar[d]\\
0\ar[r]&M'\ar[r]^{f} \ar@{=}[d]&M\ar[r]^{g} \ar[d]^{\mu} &M''\ar[r]\ar[d]^{\tau} &0\\
0\ar[r]&M'\ar[r]^{\mu'} &I\ar[r]^{\upsilon'}\ar[d]^{\upsilon} &C'\ar[r]\ar[d]^{\tau'} &0\\
&&\suspc M\ar@{=}[r]\ar[d]&\suspc M\ar[d]\\
&&0 &0.}
\end{aligned}
\end{equation}
Since $M''$ and $\suspc M$ are in ${^{\perp}\C}$, so is $C'$, which yields that the exact sequence $0 \to M' \to I \to C' \to 0$ is a special $\C$-preenvelope of $M'$. Consider the following commutative diagram with exact rows:
\begin{equation}\label{diag02}
\begin{aligned}
\xymatrix@C=20pt@R=15pt{ 0 \ar[r] & M' \ar@{=}[d]\ar[r]^{f} & M \ar[d]^{\mu}\ar[r]^{g} & M'' \ar[d]^{\tau}\ar[r] & 0 \\
0 \ar[r] & M' \ar[r]^{\mu'} \ar[d]^{f}& I \ar@{=}[d]\ar[r]^{\upsilon'}& C' \ar[d]^{\tau'}\ar[r] & 0\\
0 \ar[r] & M \ar[r]^{\mu} & I \ar[r]^{\upsilon\ \ \ }& \suspc M \ar[r] & 0. }
\end{aligned}
\end{equation}
Applying the functor $\F$ to the diagram~\eqref{diag02}, one gets the following commutative diagram with exact rows:
\begin{equation}\label{diag03}
\begin{aligned}
\xymatrix@C=30pt@R=15pt{ 0 \ar[r] & \sa_{\C}^{-1}\F(M) \ar@{.>}[d]^{\sa_{\C}^{-1}\F(f)}\ar[r]^{i} & \F(\suspc M) \ar[d]^{\F(\tau')}\ar[r]^{\ \ \F(\upsilon)} & \F(I)\ar@{=}[d] \\
0 \ar[r] & \sa_{\C}^{-1}\F(M') \ar@{.>}[d]^{\lambda}\ar[r]^{\ \ k'} & \F(C') \ar[d]^{\F(\tau)}\ar[r]^{\ \ \F(\upsilon')} & \F(I)\ar[d]^{\F(\mu)} \\
0 \ar[r] & \Kerop\F(g) \ar[r]^{k} & \F(M'') \ar[r]^{\F(g)}& \F(M),}
\end{aligned}
\end{equation}
where $\sa_{\C}^{-1}\F(f)$ and $\lambda$ are obtained by the universal property of kernels. Thus one has a homomorphism $\Upsilon_{1}= k\lambda \colon \sa_{\C}^{-1}\F(M')\to \F(M'')$, whose composition with $\F(g)$ is zero. It follows from the diagram~\eqref{diag03} that $\Upsilon_{1}\sa_{\C}^{-1}\F(f)= k\lambda\sa_{\C}^{-1}\F(f)=\F(\tau)\F(\tau')i=\F(\tau'\tau)i$=0. Thus the sequence
\[
\sa_{\C}^{-1}\F(M) \xrightarrow{\sa_{\C}^{-1}\F(f)}\sa_{\C}^{-1}\F(M') \xrightarrow[\hphantom{\sa_{\C}^{-1}\F(f)}]{\Upsilon_{1}} \F(M'') \xrightarrow[\hphantom{\sa_{\C}^{-1}\F(f)}]{\F(g)} \F(M)
\]
is a complex. In the following we prove that this sequence is actually exact.

Consider the following commutative diagram with all exact columns:
\[
\xymatrix@C=50pt@R=15pt{ &0 \ar[d]_{} &0 \ar[d]_{} &0 \ar[d]_{} &&\\
&\sa_{\C}^{-1}\F(M)\ar[d]^{i}\ar[r]^{\sa_{\C}^{-1}\F(f)} & \sa_{\C}^{-1}\F(M')\ar[d]^{k'} \ar^{\Upsilon_{1}}[r] &\F(M'')\ar@{=}[d]& \\
&\F(\suspc M)\ar[d]^{\F(\upsilon)}\ar[r]^{\F(\tau')} &\F(C')\ar[d]^{\F(\upsilon')}\ar^{\F(\tau)}[r]&\F(M'')\ar[d]& \\
&\F(I)\ar@{=}[r]&\F(I)\ar[r]&0.\\}
\]
The second non-zero row is exact as the functor $\F$ is half $\C$-exact. It follows from~\cite[III. Lemma~3.2]{CE56} that the sequence
\[
\sa_{\C}^{-1}\F(M) \xrightarrow{\sa_{\C}^{-1}\F(f)}\sa_{\C}^{-1}\F(M') \xrightarrow[\hphantom{\sa_{\C}^{-1}\F(f)}]{\Upsilon_{1}} \F(M'')
\]
is exact. Next we prove that the sequence
\[
\sa_{\C}^{-1}\F(M') \xrightarrow[\hphantom{\F(g)}]{\Upsilon_{1}}\F(M'') \xrightarrow{\F(g)} \F(M)
\]
is exact. It suffices to show $\Kerop\F(g)\subseteq \Image\Upsilon_{1}$. Consider the next pushout diagram:
\begin{equation}\label{diag04}
\begin{aligned}
\xymatrix@C=20pt@R=15pt{ &0 \ar[d]_{}& 0 \ar[d]_{} & \\
0\ar[r]^{}&M'\ar[d]^{\mu'}\ar[r]^{f}&M \ar[d]^{\varphi} \ar[r]^{g}&M''\ar@{=}[d]_{}\ar[r]^{} &0 \\
0\ar[r]^{}&I \ar[d]^{\upsilon'}\ar[r]^{\alpha\ \ \ \ \ }&I\sqcup_{M'}M\ar[d]^{\psi}\ar[r]^{\ \ \ \ \beta}&M''\ar[r]^{}&0 \\
&C' \ar[d]_{} \ar@{=}[r]_{} &C' \ar[d]_{} & &\\
& 0 & 0. && }
\end{aligned}
\end{equation}
Define $\delta \colon M''\to I\sqcup_{M'}M$ by $\delta(m'')=\bracks[\big]{\parens[\big]{-\mu(m),m}}_{\Image(-\mu',f)}$, where $m$ is in $M$ such that $g(m)=m''$. Then it is clear that $\delta$ is a homomorphism such that $\beta\delta=\idop_{M''}$. Moreover, by the diagram~\eqref{diag02} one has $\psi\delta(m'')=\upsilon'\parens[\big]{-\mu(m)}=-\tau g(m)=-\tau(m'')$, which yields that $\psi\delta=-\tau$. We are now ready to prove $\Kerop\F(g)\subseteq \Image\Upsilon_{1}$. For each element $a\in\F(M'')$ such that $\F(g)(a)=0$. By the diagram~\eqref{diag04} one has $0=\F(g)(a)=\F(\beta\varphi)(a)=\F(\varphi)\F(\beta)(a)$, it follows that $\F(\beta)(a)\in\Kerop\F(\varphi)=\Image\F(\psi)$, as the sequence
\[
\F(C')\xrightarrow{\F(\psi)}\F(I\sqcup_{M'}M)\xrightarrow{\F(\varphi)}\F(M)
\]
is exact. So there is an element $b\in \F(C')$ such that $\F(\psi)(b)=\F(\beta)(a)$. Furthermore, one has $\F(\delta)\F(\psi)(b)=\F(\delta)\F(\beta)(a)$, which yields $a=-\F(\tau)(b)$ as $\beta\delta=\id[M'']$ and $\psi\delta=-\tau$. Since $\F(\upsilon')(b)=\F(\psi\alpha)(b)=\F(\alpha)\F(\psi)(b)=\F(\alpha)\F(\beta)(a)=0$, one has $b\in \Kerop\F(\upsilon')=\Image k'$ by diagram~\eqref{diag03}, it follows that there is an element $c\in \sa_{\C}^{-1}\F(M')$ satisfying $b=k'(c)$. Consequently, $a=-\F(\tau)(b)=-\F(\tau)\parens[\big]{k'(c)}=-\Upsilon_{1}(c)\in \Image\Upsilon_{1}$, so one obtains $\Kerop\F(g)\subseteq \Image\Upsilon_{1},$ as desired. Hence one gets an exact sequence
\[
\sa_{\C}^{-1}\F(M)\xrightarrow{\sa_{\C}^{-1}\F(f)} \sa_{\C}^{-1}\F(M')\xrightarrow[\hphantom{\sa_{\C}^{-1}\F(f)}]{\Upsilon_{1}} \F(M'')\xrightarrow[\hphantom{\sa_{\C}^{-1}\F(f)}]{\F(g)} \F(M),
\]
which yields that the sequence
\[
\sa_{\C}^{-1}\F(M)\xrightarrow{\sa_{\C}^{-1}\F(f)} \sa_{\C}^{-1}\F(M')\xrightarrow[\hphantom{\sa_{\C}^{-1}\F(f)}]{\Upsilon_{1}} \F(M'')\xrightarrow[\hphantom{\sa_{\C}^{-1}\F(f)}]{\F(g)} \F(M)\xrightarrow[\hphantom{\sa_{\C}^{-1}\F(f)}]{\F(f)} \F(M')
\]
is exact as $\F$ is half $\C$-exact. We notice that $\sa_{\C}^{-i}\F(-)$ is half $\C$-exact for each $i\geq1$ by Lemma~\ref{half-exact}, then one gets the exact sequence in the statement.
\end{proof}

\begin{enonce*}[remark]{Connected (cohomology) sequences}%]\label{5.1}
A family $\F^{\ast}=\braces[\big]{\F^{n} \st[\big] n\in\ZZ}$ of contravariant additive functors from $\RMod$ to $\Ab$ is called a $\C$-\emph{connected sequence} if for each short exact sequence $0 \to A' \to A \to A'' \to 0$ of $R$-modules with $A''\in{^{\perp}\C}$, the sequence
\begin{equation}\label{1}
\dotsb \to\F^{n}(A'')\to\F^{n}(A)\to\F^{n}(A')\xra{\delta}
\F^{n+1}(A'')\to \dotsb
\end{equation}
of abelian groups is a complex. If furthermore the sequence~\eqref{1} is exact, then $\F^{\ast}$ is called a $\C$-\emph{cohomology sequence}.

Dually, one has the definition of $\D$-\emph{connected/cohomology} sequences.
\end{enonce*}

Mislin and Nucinkis introduced the concepts of $P$-completions and $I$-completions of cohomology sequence of functors in~\cite{GM94} and~\cite{BN98}, respectively. In the following we introduce the relative versions of these concepts.

\begin{defi}\label{Universal property C}
Let\/ $\F^{*} =\braces[\big]{\F^{n} \st[\big] n\in\ZZ}$ be a $\C$-cohomology sequence of contravariant functors. The $\C$-\emph{completion} of\/ $\F^{*}$ is a $\C$-cohomology sequence\/ $\TF{*} = \braces[\big]{\TF{n} \st[\big] n\in\ZZ}$ of contravariant functors together with a morphism $\tau \colon \F^{*}\to\TF{*}$ satisfying the following conditions:
\begin{itemize}
\item $\TF{n}(C)=0$ for each $R$-module $C\in\C$ and all $n\in\ZZ$;
\item if $\T^{*} =\braces[\big]{\T^{n} \st[\big] n\in\ZZ}$ is a $\C$-cohomology sequence of contravariant functors satisfying $\T^{n}(C)=0$ for each $C\in\C$ and all $n\in\ZZ$, and if $\nu \colon \F^{*}\to\T^{*}$ is a morphism, then there exists a unique morphism $\sigma \colon \TF{*}\to\T^{*}$ such that $\sigma\tau = \nu$.
\end{itemize}
Similarly, for a $\D$-cohomology sequence $\G^{*} =\braces[\big]{\G^{n} \st[\big] n\in\ZZ}$ of covariant functors one has the definition of\/ $\D$-\emph{completion} $\varsigma \colon \G^{*}\to \TG{*}$.
\end{defi}

The next lemma is proved similarly as in~\cite[Proposition~2.3]{BN98}. For the sake of readability, we provide its proof here.

\begin{lemm}\label{lem1}
Let\/ $\F^{\leq0}=\braces[\big]{\F^{n} \st[\big] n\leq0}$ and\/ $\T^{\leq0}=\braces[\big]{\T^{n} \st[\big] n\leq0}$ be $\C$-connected sequences of contravariant functors and $\phi^{0} \colon \F^{0}\to \T^{0}$ a natural transformation. If\/ $\T^{\leq0}$ is a $\C$-cohomology sequence and satisfies $\T^{-m}(C) = 0$ for each $R$-module $C\in \C$ and all $m>0$, then the following statements hold:
\begin{enumerate}\alphenumi
\item \label{lemm_9_a} $\phi^{0}$ extends uniquely to $\phi^{\leq0} \colon \F^{\leq0}\to \T^{\leq0}$ and $\phi^{\leq0}$ factors uniquely through the canonical morphism $\F^{\leq0}\to \sa_{\C}^{\leq0}\F^{0}$;
\item \label{lemm_9_b} if\/ $\F^{0}$ is half\/ $\C$-exact and $\phi^{0}$ is an equivalence, then the induced morphism $\sa_{\C}^{\leq0}\F^{0}\to \T^{\leq0}$ is an equivalence.
\end{enumerate}
\end{lemm}

\begin{proof}
\begin{proof}[(\ref{lemm_9_a})]
We proceed by induction on $n$. Suppose that natural transformations $\phi^{q} \colon \F^{q}\to \T^{q}$ are already defined for $n<q\leq 0$, and commute with the connecting homomorphisms. Let $M$ be an $R$-module and $0 \to M \to I \to \suspc M \to 0$ a special $\C$-preenvelope. Then one gets the following commutative diagram:
\[
\xymatrix@C=20pt@R=15pt{ & \F^{n}(M) \ar@{.>}[d]^{\phi^{n}_{M}}\ar[r] & \F^{n+1}(\suspc M) \ar[d]_{}\ar[r]^{} & \F^{n+1}(I) \ar[d]_{} \\
0 \ar[r] & \T^{n}(M) \ar[r] & \T^{n+1}(\suspc M) \ar[r]^{} & \T^{n+1}(I),}
\]
where the top row is a complex and the bottom row is exact and $\phi^{n}_{M}$ is obtained by the universal property of kernels. Moreover, it is easy to check that $\phi^{n} \colon \F^{n}\to \T^{n}$ is a natural transformation.

Now we prove that $\phi^{n}$ commutes with the connecting homomorphisms. Let $0 \to M' \to M \to M'' \to 0$ be an exact sequence of $R$-modules with $M''\in{^{\perp}\C}$ and $0 \to M' \to I' \to \suspc M' \to 0$ a special $\C$-preenvelope. Consider the following commutative diagram with exact rows
\[
\xymatrix@C=20pt@R=15pt{ 0 \ar[r] & M' \ar@{=}[d]^{}\ar[r] & M \ar[d]_{}\ar[r]^{} & M'' \ar[d]_{}\ar[r]^{}& 0 \\
0 \ar[r] & M' \ar[r] & I' \ar[r]^{} & \suspc M'\ar[r]^{}& 0.}
\]
This yields a commutative diagram:
\[
\xymatrix@C=20pt@R=15pt{ &\F^{n}(M') \ar[d]^{}\ar[r] & \F^{n+1}(\suspc M') \ar[d]_{}\ar[r]^{} & \F^{n+1}(M'') \ar[d]_{} \\
&\T^{n}(M') \ar[r] & \T^{n+1}(\suspc M') \ar[r]^{} & \T^{n+1}(M'').}
\]
Hence the desired commutativity relation follows.

In the following we prove that $\phi^{\leq0} \colon \F^{\leq0}\to \T^{\leq0}$ factors uniquely through the canonical morphism $\delta^{\leq0} \colon \F^{\leq0}\to \sa_{\C}^{\leq0}\F^{0}$. It suffices to show that $\phi^{-1} \colon \F^{-1}\to \T^{-1}$ factors uniquely through the canonical morphism $\delta^{-1} \colon \F^{-1}\to \sa_{\C}^{-1}\F^{0}$. For each $R$-module $M$ and a special $\C$-preenvelope $0 \to M \to I \to \suspc M \to 0$, one has the following commutative diagram with exact rows:
\[
\xymatrix@C=20pt@R=15pt{ 0 \ar[r] & \sa_{\C}^{-1}\F^{0}(M) \ar@{.>}[d]^{\lambda^{-1}_{M}}\ar[r] & \F^{0}(\suspc M) \ar[d]_{}\ar[r]^{} & \F^{0}(I) \ar[d]_{} \\
0 \ar[r] & \T^{-1}(M) \ar[r] & \T^{0}(\suspc M) \ar[r]^{} & \T^{0}(I),}
\]
where $\lambda^{-1}_{M}$ is obtained by the universal property of kernels. Similarly, it is easy to check that $\lambda^{-1} \colon \F^{-1}\to \sa_{\C}^{-1}\F^{0}$ is a natural transformation and satisfies $\varphi^{-1}=\lambda^{-1}\delta^{-1}$. Finally, the conclusion follows by induction.
\let\qed\relax
\end{proof}
\begin{proof}[(\ref{lemm_9_b})] It follows directly from Proposition~\ref{complex sequence} and the Five Lemma in view of the special $\C$-preenvelope $0 \to M \to I \to \suspc M \to 0$ of an $R$-module $M$.
\end{proof}
\let\qed\relax
\end{proof}
Dually, we have the following result that can be found in~\cite[Theorem~1.1]{GM94}.

\begin{lemm}
Let\/ $\G^{\leq0}=\braces[\big]{\G^{n} \st[\big] n\leq0}$ and $\U^{\leq0}=\braces[\big]{\U^{n} \st[\big] n\leq0}$ be\/ $\D$-connected sequences of covariant functors and $\varphi^{0} \colon \G^{0}\to \U^{0}$ a natural transformation. If\/ $\G^{\leq0}$ is a $\D$-cohomology sequence and satisfies\/ $\G^{-m}(D) = 0$ for each $R$-module $D\in \D$ and all $m>0$, then the following statements hold:
\begin{enumerate}\alphenumi
\item $\varphi^{0}$ extends uniquely to $\varphi^{\leq0} \colon \G^{\leq0}\to \U^{\leq0}$ and $\varphi^{\leq0}$ factors uniquely through the canonical morphism\/ $\G^{\leq0}\to \sa_{\D}^{\leq0}\G^{0}$;
\item if\/ $\G^{0}$ is half\/ $\D$-exact and $\varphi^{0}$ is an equivalence, then the induced morphism\/ $\sa_{\D}^{\leq0}\G^{0}\to \U^{\leq0}$ is an equivalence.
\end{enumerate}
\end{lemm}

\begin{remark}[Construction]\label{A4}
We fix $n\in \ZZ$. Let $\F^{\ast} =\braces[\big]{\F^{n} \st[\big] n\in\ZZ}$ be a $\C$-connected sequence of contravariant functors and $M$ an $R$-module. For each $k\geq 1$, the exact sequence $0 \to \suspc^{k}M \to I^{k} \to \suspc^{k+1}M \to 0$ yields a complex
\[
\F^{n+k}(\suspc^{k}M) \xra{\delta} \F^{n+k+1}(\suspc^{k+1}M) \to\F^{n+k+1}(I^k).
\]
The connecting homomorphism $\delta$ induces a homomorphism from $\F^{n+k}(\suspc^{k}M)$ to the kernel $\sa_{\C}^{-1}\F^{n+k+1}(\suspc^{k}M)\cong\sa_{\C}^{-(k+1)}\F^{n+k+1}(M)$; see Remark~\ref{A2}. Composed with the natural embedding from $\sa_{\C}^{-1}\F^{n+k}(\suspc^{k-1}M)\cong{\sa_{\C}^{-k}}\F^{n+k}(M)$ to $\F^{n+k}(\suspc^{k}M)$, it yields a homomorphism
\[
\delta \colon \sa_{\C}^{-k}\F^{n+k}(M) \to \sa_{\C}^{-(k+1)}\F^{n+k+1}(M).
\]
Similarly, for a $\D$-connected sequence of covariant functors $\G^{\ast} =\braces[\big]{\G^{n} \st[\big] n\in\ZZ}$, one has a homomorphism
\[
\partial \colon \sa_{\D}^{-k}\G^{n+k}(M) \to \sa_{\D}^{-(k+1)}\G^{n+k+1}(M)
\]
for each $R$-module $M$ and any $k\geq 1$.
\end{remark}

The following result is proved similarly as in~\cite[Theorem~2.2]{GM94} (see also~\cite[Theorem~2.5]{BN98}).

\begin{theo}\label{Mislin}
The following statements hold.
\begin{enumerate}\alphenumi
\item \label{theo_12_a} Each $\C$-cohomology sequence $\F^{*} =\braces[\big]{\F^{n} \st[\big] n\in\ZZ}$ of contravariant functors admits a unique $\C$-completion $\TF{*} = \braces[\big]{\TF{n} \st[\big] n\in\ZZ}$ with
\[
\TF{n}(M)=\colim_{i}\sa^{-i}_{\C}\F^{n+i}(M)
\]
for each $R$-module $M$.
\item \label{theo_12_b} Each $\D$-cohomology sequence $\G^{*} =\braces[\big]{\G^{n} \st[\big] n\in\ZZ}$ of covariant functors admits a unique $\D$-completion $\TG{\ast} = \braces[\big]{\TG{n} \st[\big] n\in\ZZ}$ with
\[
\TG{n}(N)=\colim_{j}\sa^{-j}_{\D}\G^{n+j}(N)
\]
for each $R$-module $N$.
\end{enumerate} Here, the homomorphisms in the direct systems are provided in~Remark~\ref{A4}.
\end{theo}

\begin{proof}
We only prove~\eqref{theo_12_a}; the statement~\eqref{theo_12_b} is proved similarly.

Since $\F^{*}$ is a $\C$-cohomology sequence of contravariant functors, one gets a $\C$-cohomology sequence $\sa_{\C}^{\leq0}\F^{n}$ by Proposition~\ref{complex sequence} for each $n\in\ZZ$, and it can be extended to a $\C$-cohomology sequence $\F^{*}\langle n\rangle$ by setting
\[
\F^{j}\langle n\rangle =
\begin{cases}
\sa_{\C}^{j-n}\F^{n}, 	& \text{if $j<n$}, 		\\
\F^{n}, 					& \text{if $j\geq n$}.
\end{cases}
\]
Then the identity morphism $\F^{n}\to \F^{n}$ induces a unique morphism $\F^{\leq n}\to \sa_{\C}^{\leq0}\F^{n}$ and we extend it further to $\tau^{*}_{n} \colon \F^{*}\to \F^{*}\langle n\rangle$ with $\tau^{j}_{n} = \idop_{\F^{j}}$ for all $j\geq n$. Similarly, for all $m\geq n$, the identity morphism $\F^{m}\to \F^{m}$ extends uniquely to a morphism $\tau^{*}_{n,m} \colon \F^{*}\langle n\rangle \to \F^{*}\langle m\rangle$ with $\tau^{j}_{n,m} = \idop_{\F^{j}}$ for all $j\geq m$. Hence we now define
\[
\TF{*} = \colim\braces[\big]{\F^{*}\langle n\rangle \st[\big] \tau^{*}_{n,m}}.
\]
Since $\tau^{*}_{n,m}\tau^{*}_{n} = \tau^{*}_{m}$ for $m\geq n$, there is a natural morphism
\[
\tau^{*} = \colim\tau^{*}_{n} \colon \F^{*}\to \TF{*}.
\]
The exactness of colimits implies that $\TF{*}$ is a $\C$-cohomology sequence. So for each $R$-module $M$, one has
\[
\TF{j}(M) = \colim_{i}\sa_{\C}^{-i}\F^{i+j}(M), \ \ \ i\geq0
\]
such that $\TF{j}(C) = 0$ for each $R$-module $C\in \C$ and all $j\in\ZZ$, as $\sa_{\C}^{-i}\F^{i+j}(C) = 0$ for $i>0$. Next, we show the universal property of $\tau^{*}$. Let $\F^{*}\to \T^{*}$ be any morphism with $\T^{*}$ a $\C$-cohomological sequence of contravariant functors and $\T^n(C)=0$ for each $R$-module $C\in \C$ and any $n\in\ZZ$. Then each $\F^{n}\to \T^{n}$ extends uniquely to $\sa_{\C}^{\leq0}\F^{n} \to \sa_{\C}^{\leq0}\T^{n}$, and one has $\sa_{\C}^{\leq0}\T^{n}\cong \T^{\leq n}$ for the identity morphism $\T^{n}\to \T^{n}$ by Lemma~\ref{lem1}. Consequently, one gets a unique morphism $\TF{*}\to \T^{*}$ and the morphism $\F^{*}\to \T^{*}$ factors uniquely through $\tau^{*}$. The uniqueness of the $\C$-completion is immediately obtained by its definition.
\end{proof}

The next corollary is immediate by the uniqueness of completion.

\begin{coro}\label{4.5}
The following statements hold.
\begin{enumerate}\alphenumi
\item If\/ $\F^{*}$ is a $\C$-cohomology sequence of contravariant functors satisfying $\F^{n}(C)=0$ for each $C\in\C$ and all $n\in\ZZ$, then there is a natural isomorphism $\F^{*}\cong\TF{*}$.
\item If\/ $\G^{*}$ is a $\D$-cohomology sequence of covariant functors satisfying $\G^{n}(D)=0$ for each $D\in\D$ and all $n\in\ZZ$, then there is a natural isomorphism $\G^{*}\cong\TG{*}$.
\end{enumerate}
\end{coro}

\begin{rema}\label{complete and mislin}
Let $M$ and $N$ be $R$-modules, and let $\F^{\ast}=\Ext{\ast}{-}{N}$ and $\G^{*}=\Ext{\ast}{M}{-}$. It follows from Theorem~\ref{Mislin} that the $\C$-completion of $\F^{*}$ and $\D$-completion of $\G^{*}$ are
\[
\TF{\ast}=\braces[\big]{\TF{n} = \colim_{i}\sa_{\C}^{-i}\Ext{n+i}{-}{N} \st[\big] n\in\mathbb{Z}}
\]
and
\[
\TG{\ast}=\braces[\big]{\TG{n}=\colim_{j}\sa_{\D}^{-j}\Ext{n+j}{M}{-} \st[\big] n\in\mathbb{Z}},
\]
respectively. In~\cite[Definition~3.4]{GL22}, we defined a contravariant additive functor $\Text[\C]{n}{-}{N}$ as
\[
\Text[\C]{n}{M}{N}=\colim_{i}\CHom{\suspc^{i}M}{\suspc^{i+n}N}
\]
for each $R$-module $M$; it is actually the $n$-th \emph{complete cohomology} of $M$ and $N$ with respect to $\C$. We also defined in~\cite[Definition~3.7]{GL22} a covariant additive functor $\Cext[\D]{n}{M}{-}$ as
\[
\Cext[\D]{n}{M}{N}=\colim_{i}\DHom{\Omega^{\D}_{i+n}M} {\Omega^{\D}_{i}N}
\]
for each $R$-module $N$; it is actually the $n$-th \emph{complete cohomology} of $M$ and $N$ with respect to $\D$. If $\Ext{\geq 1}{{^{\perp}\C}}{\C}$ = 0 = $\Ext{\geq 1}{\D}{\D^{\bot}}$, then it follows from~\cite[Proposition~3.12]{GL22} that the functors $\Text[\C]{\ast}{-}{N}$ and $\Cext[\D]{\ast}{M}{-}$ are actually the $\C$-completion of $\Ext{*}{-}{N}$ and the $\D$-completion of $\Ext{\ast}{M}{-}$, respectively.
\end{rema}

\section{Balancedness of relative complete cohomology}\label{balancedness}
In this section, we study the relation between the complete cohomology groups $\Text[\C]{n}{M}{N}$ and $\Cext[\D]{n}{M}{N}$ given in Remark~\ref{complete and mislin}.

\begin{enonce*}[remark]{Balanced pair}%]\label{banlance pair}
Following~\cite{Chen}, a pair $(\x, \y)$ of subcategories of $\RMod$ is called a \emph{balanced pair} if the following conditions hold:
\begin{itemize}
\item $\x$ is precovering and $\y$ is preenveloping;
\item for each $R$-module $M$, there is a proper $\x$-resolution $P\to M$ such that it is $\Hom{-}{\y}$-exact;
\item for each $R$-module $N$, there is a proper $\y$-coresolution $N\to I$ such that it is $\Hom{\x}{-}$-exact.
\end{itemize}
\end{enonce*}

\begin{rema}\label{balance of relative cohomology}
If $(\x, \y)$ is a balanced pair in $\RMod$, then by~\cite[Lemma~2.1]{Chen} there is a natural isomorphism $\Ext[\x R]{n}{M}{N}\cong\Ext[R\y]{n}{M}{N}$ for all $R$-modules $M$ and $N$, and each $n\geq0$.
\end{rema}

\begin{lemm}\label{4.9}
Let $(\D,\C)$ be a balanced pair. Then for all $R$-modules $M$ and $N$ the following statements hold.
\begin{enumerate}\alphenumi
\item \label{lemm_16_a} The family\/ $\Text[\C]{*}{M}{-}$ is a\/ $\D$-cohomology sequence of covariant functors.
\item \label{lemm_16_b} The family\/ $\Cext[\D]{*}{-}{N}$ is a\/ $\C$-cohomology sequence of contravariant functors.
\end{enumerate}
\end{lemm}
\begin{proof}
We only prove~\eqref{lemm_16_a}; the statement~\eqref{lemm_16_b} is proved similarly.

Let $0 \to N' \to N \to N'' \to 0$ be a short exact sequence of $R$-modules with $N'\in\D^{\bot}$. Then it is $\Hom{\D}{-}$-exact. Since $(\D,\C)$ is a balanced pair, by~\cite[Proposition~2.2]{Chen} the above sequence is also $\Hom{-}{\C}$-exact. Thus by~\cite[Proposition~4.7]{GL22} the sequence
\[
\dotsb \to \Text[\C]{n}{M}{N'} \to \Text[\C]{n}{M}{N} \to \Text[\C]{n}{M}{N''} \to \Text[\C]{n+1}{M}{N'} \to \dotsb
\]
is exact. This yields that $\Text[\C]{*}{M}{-}$ is a $\D$-cohomology sequence.
\end{proof}

Let $M$ and $N$ be $R$-modules. For the functor $\F=\Ext[R]{i}{-}{N}$, the value of the left satellite functor ${\sa^{-n}_{\C}}\F$ at $M$, ${\sa^{-n}_{\C}}\F(M)$, is always denoted ${\sa^{-n}_{\C}}\Ext[R]{i}{M}{N}$. For the functor $\G=\Ext[R]{i}{M}{-}$, the value of the left satellite functor ${\sa^{-n}_{\D}}\G$ at $N$, ${\sa^{-n}_{\D}}\G(N)$, is always denoted ${\sa^{-n}_{\D}}\Ext[R]{i}{M}{N}$. Thus $\sa^{-j}_{\D}\sa^{-i}_{\C}\Extop^{n}_{R}(M,N)$ denotes $\sa^{-j}_{\D}\parens[\big]{\sa^{-i}_{\C}\Extop^{n}_{R}(M,-)}(N)$ and $\sa^{-i}_{\C}\sa^{-j}_{\D}\Extop^{n}_{R}(M,N)$ denotes $\sa^{-i}_{\C}\parens[\big]{\sa^{-j}_{\D}\Extop^{n}_{R}(-,N)}(M)$.

The next result is a relative version of~\cite[III. Theorem~7.1]{CE56}.

\begin{lemm}\label{CE}
For all $R$-modules $M$ and $N$, and each $i, j, n\geq0$, there is a natural isomorphism
\[
\sa^{-j}_{\D}\sa^{-i}_{\C}\Extop^{n}_{R}(M,N) \cong \sa^{-i}_{\C}\sa^{-j}_{\D}\Extop^{n}_{R}(M,N).
\]
\end{lemm}

\begin{proof}
It is enough to prove $\sa^{-1}_{\D}\sa^{-1}_{\C}\Extop^{n}_{R}(M,N) \cong \sa^{-1}_{\C}\sa^{-1}_{\D}\Extop^{n}_{R}(M,N)$ for each $n\geq0$. Let $0\to M\to I\to \suspc M\to 0$ be a special $\C$-preenvelope of $M$ and $0\to \Omega^{\D}N \to J\to N \to 0$ a special $\D$-precover of $N$. Then one obtains the following two commutative diagrams with exact rows and columns:
\begin{equation}\label{diag05}
\begin{aligned}
\xymatrix@C=12pt@R=15pt{ &0 \ar[d]_{}\\
&\sa^{-1}_{\C}\sa^{-1}_{\D}\Extop^{n}_{R}(M,N)\ar[d]_{}\\
0\ar[r]&\sa^{-1}_{\D}\Extop^{n}_{R}(\suspc M,N) \ar[d]\ar[r]^{} & \Extop^{n}_{R}(\suspc M,\Omega^{\D}N) \ar[d]_{}\ar[r]^{} & \Extop^{n}_{R}(\suspc M,J) & \\
0\ar[r]&\sa^{-1}_{\D}\Extop^{n}_{R}(I,N) \ar[r]^{} & \Extop^{n}_{R}(I,\Omega^{\D}N), & & }
\end{aligned}
\end{equation}
\begin{equation}\label{diag06}
\begin{aligned}
\xymatrix@C=12pt@R=15pt{ &0 \ar[d]_{}\\
&\sa^{-1}_{\D}\sa^{-1}_{\C}\Extop^{n}_{R}(M,N)\ar[d]_{}\\
0\ar[r]&\sa^{-1}_{\C}\Extop^{n}_{R}(M,\Omega^{\D}N) \ar[d]\ar[r]^{} & \Extop^{n}_{R}(\suspc M,\Omega^{\D}N) \ar[d]_{}\ar[r]^{} & \Extop^{n}_{R}(I,\Omega^{\D}N) & \\
0\ar[r]&\sa^{-1}_{\C}\Extop^{n}_{R}(M,J) \ar[r]^{} & \Extop^{n}_{R}(\suspc M,J). & & }
\end{aligned}
\end{equation}
The diagrams~\eqref{diag05} and~\eqref{diag06} imply that both the sequences
\[
0\to\sa^{-1}_{\C}\sa^{-1}_{\D}\Extop^{n}_{R}(M,N)\to\Extop^{n}_{R}(\suspc M,\Omega^{\D}N) \to\Extop^{n}_{R}(I,\Omega^{\D}N)\oplus\Extop^{n}_{R}(\suspc M,J)
\]
and
\[
0\to\sa^{-1}_{\D}\sa^{-1}_{\C}\Extop^{n}_{R}(M,N)\to\Extop^{n}_{R}(\suspc M,\Omega^{\D}N) \to\Extop^{n}_{R}(I,\Omega^{\D}N)\oplus\Extop^{n}_{R}(\suspc M,J)
\]
are exact. Thus one gets the following commutative diagram with exact rows:
\[
\xymatrix@C=6pt@R=12pt{
0 \ar[r] & \sa^{-1}_{\C}\sa^{-1}_{\D}\Extop^{n}_{R}(M,N) \ar@{.>}[d]^{\phi}\ar[r]^{} & \Extop^{n}_{R}(\suspc M,\Omega^{\D}N) \ar@{=}[d]\ar[r]^{} & \Extop^{n}_{R}(I,\Omega^{\D}N)\oplus\Extop^{n}_{R}(\suspc M,J) \ar@{=}[d] & \\
0 \ar[r] & \sa^{-1}_{\D}\sa^{-1}_{\C}\Extop^{n}_{R}(M,N) \ar[r]^{} & \Extop^{n}_{R}(\suspc M,\Omega^{\D}N) \ar[r]^{}& \Extop^{n}_{R}(I,\Omega^{\D}N)\oplus\Extop^{n}_{R}(\suspc M,J), & }
\]
where $\phi$ is obtained by the universal property of kernels, moreover $\phi$ is an isomorphism by the Five Lemma, as desired.
\end{proof}

It is easy to see that $\braces[\big]{\colim_{j}\sa^{-j}_{\D}\Extop^{n+j}_{R}(-,N) \st[\big] n\in\ZZ}$ is a $\C$-connected sequence of contravariant functors, and $\braces[\big]{\colim_{i}\sa^{-i}_{\C}\Extop^{n+i}_{R}(M,-) \st[\big] n\in\ZZ}$ is a $\D$-connected sequence of covariant functors, where the morphisms in the direct systems are provided by those $\delta$ and $\partial$ in~Remark~\ref{A4}.

\begin{lemm}\label{lem5.1}
Let $M$ and $N$ be $R$-modules. For each $n\in\ZZ$, there exists a natural isomorphism
\[
\colim_{j}\sa^{-j}_{\D}\colim_{i}\sa^{-i}_{\C}
\Extop^{n+i+j}_{R}(M,N)\cong\colim_{i}\sa^{-i}_{\C}\colim_{j}\sa^{-j}_{\D} \Extop^{n+i+j}_{R}(M,N).
\]
\end{lemm}
\begin{proof}
By~Remark~\ref{A4} and Lemma~\ref{CE} we have the following diagram:
\begin{equation}\label{2}
\begin{aligned}
\scalebox{0.9}[0.9]{\xymatrixcolsep{1pc}\xymatrix@C=12pt@R=12pt{
\Extop^{n}_{R}(M,N)\ar[r]^{\delta\ \ \ \ \ }\ar[d]^{\partial} &\sa^{-1}_{\C}\Extop^{n+1}_{R}(M,N)\ar[d]^{\partial}\ar[r]^{\delta} &\sa^{-2}_{\C}\Extop^{n+2}_{R}(M,N)\ar[d]^{\partial}\ar[r]^{} &\dotsb\\
\sa^{-1}_{\D}\Extop^{n+1}_{R}(M,N)\ar[r]^{\delta\ \ \ }\ar[d]^{\partial} &\sa^{-1}_{\C}\sa^{-1}_{\D}\Extop^{n+2}_{R}(M,N)\ar[r]^{\delta}\ar[d]^{\partial}&
\sa^{-2}_{\C}\sa^{-1}_{\D}\Extop^{n+3}_{R}(M,N)\ar[d]^{\partial} \ar[r]^{} &\dotsb\\
\sa^{-2}_{\D}\Extop^{n+2}_{R}(M,N)\ar[r]^{\delta\ \ } \ar[d]^{} &\sa^{-1}_{\C}\sa^{-2}_{\D}\Extop^{n+3}_{R}(M,N)\ar[r]^{\delta}\ar[d]^{}&
\sa^{-2}_{\C}\sa^{-2}_{\D}\Extop^{n+4}_{R}(M,N)\ar[r]^{}\ar[d]^{} &\dotsb \\
\vdots &\vdots &\vdots}}
\end{aligned}
\end{equation}

\begin{enonce*}{Claim}
Each of the squares in the above diagram is anticommutative.
\end{enonce*}

\begin{proof}[Proof of the claim]
We only prove the anticommutativity for the square in the upper left corner; one may prove similarly for the others.

From the constructions of $\delta$ and $\partial$ in~Remark~\ref{A4} and the definition of left satellite functors, one gets the next $3$D-diagram with side squares all commutative, as the connected homomorphisms are natural.
\[
\xymatrix@C=0pt@R=15pt{ &\Extop^{n}_{R}(M,N) \ar[rrrrrr]\ar[dl] \ar@{>.>}[dd] & && &&& \sa^{-1}_{\C}\Extop^{n+1}_{R}(M, N) \ar[dl]\ar@{>->}[dd] \\
\sa^{-1}_{\D}\Extop^{n+1}_{R}(M,N) \ar[rrrrrr]\ar@{>->}[dd] & & & &&& \sa^{-1}_{\C}\sa^{-1}_{\D}\Extop^{n+2}_{R}(M,N) \ar@{>->}[dd] \\
&\Extop^{n}_{R}(M,N) \ar@{.>}[rrrrrr]\ar@{.>}[dl] & & & &&&\Extop^{n+1}_{R}(\suspc M,N)\ar[dl] & \\
\Extop^{n+1}_{R}(M,\Omega^{\D}N) \ar[rrrrrr] & & & &&&\Extop^{n+2}_{R}(\suspc M,\Omega^{\D}N).}
\]
In the above diagram, the bottom square is anticommutative; see Rotman~\cite[Theorem~11.24]{JR79}. Thus the top square is anticommutative, as all the vertical homomorphisms are natural embeddings. This finishes the proof of the claim.
\let\qed\relax
\end{proof}

Thus the diagram~\eqref{2} can be rewritten as the next commutative diagram:
\[
\xymatrix@C=20pt@R=15pt{
\Ext{n}{M}{N} \ar[d]^{\partial\partial} \ar[r]^{\delta\delta\ \ \ } & \sa^{-2}_{\C}\Extop^{n+2}_{R}(M,N) \ar[d]^{\partial\partial} \ar[r] & \dotsb \\
\sa^{-2}_{\D}\Extop^{n+2}_{R}(M,N) \ar[d] \ar[r]^{\delta\delta\ \ \ } & \sa^{-2}_{\C}\sa^{-2}_{\D}\Extop^{n+4}_{R}(M,N) \ar[d] \ar[r] & \dotsb \\
\vdots & \vdots & }
\]
Now the desired isomorphism in the statement follows.
\end{proof}

%\begin{ipg}\label{4.7}
Assume that $(\D,\C)$ is a balanced pair and ${\Ext{\geq1}{{^{\perp}\C}}{\C}}=0=\Ext{\geq1}{\D}{\D^{\bot}}$. By~\cite[Proposition~3.12]{GL22} one has
\[
\Cext[\D]{n}{M}{N}\cong\colim_{j}\sa^{-j}_{\D}\Ext[R]{n+j}{M}{N}
\]
for all $R$-modules $M$ and $N$ and each $n\in\ZZ$. From Lemma~\ref{4.9}, the family
\[
\F(N)^{*}=\braces[\big]{\F(N)^{n}=\Cext[\D]{n}{-}{N} \st[\big] n\in\ZZ}
\]
is a $\C$-cohomology sequence of contravariant functors. Hence by Theorem~\ref{Mislin} it admits a $\C$-completion $\tF{*}$ with
\[
\tF{n}(M)=\colim_{i}\sa^{-i}_{\C}\F(N)^{n+i}(M)=\colim_{i}\sa^{-i}_{\C} \colim_{j}\sa^{-j}_{\D}\Extop^{n+i+j}_{R}(M,N).
\]

Similarly, by~\cite[Proposition~3.12]{GL22} one has
\[
\Text[\C]{n}{M}{N}\cong\colim_{i}\sa^{-i}_{\C}\Ext{n+i}{M}{N}
\]
for all $R$-modules $M$ and $N$ and each $n\in\ZZ$. From Lemma~\ref{4.9}, the family
\[
\G(M)^{*}=\braces[\big]{\G(M)^{n}=\Text[\C]{n}{M}{-} \st[\big] n\in\ZZ}
\]
is a $\D$-cohomology sequence of covariant functors, and so by Theorem~\ref{Mislin} it admits a $\D$-completion $\tG{*}$ with
\[
\tG{n}(N)=\colim_{j}\sa^{-j}_{\D}\G(M)^{n+j}(N)=\colim_{j}\sa^{-j}_{\D} \colim_{i}\sa^{-i}_{\C}\Extop^{n+j+i}_{R}(M,N).
\]
Therefore, it follows from Lemma~\ref{lem5.1} that there is a natural isomorphism
\begin{equation}\label{isomorphism}
\tF{n}(M)\cong\tG{n}(N)
\end{equation}
for all $R$-modules $M$ and $N$, and each $n\in\ZZ$.
%\end{ipg}
Combining the vanishing results in~\cite[Theorems~4.9 and~4.11]{GL22}, we give a balanced result for complete cohomology groups $\Text[\C]{n}{M}{N}$ and $\Cext[\D]{n}{M}{N}$.

\begin{theo}\label{balance}
Let $(\D,\C)$ be a balanced pair. Assume that $\D$ and $\C$ are closed under direct summands and\/ ${\Ext{\geq1}{\D}{\D^{\bot}}}=0={\Ext{\geq1}{{^{\perp}\C}}{\C}}$. Then the following statements are equivalent.
\begin{enumerate}\romanenumi
\item \label{theo_19_i} All $R$-modules in $\C$ have finite $\D$-projective dimension and all $R$-modules in $\D$ have finite $\C$-injective dimension.
\item \label{theo_19_ii} For all $R$-modules $M$ and $N$ and each $n\in\ZZ$, there is a natural isomorphism
\[
\Text[\C]{n}{M}{N} \cong \Cext[\D]{n}{M}{N}.
\]
\item \label{theo_19_iii} For each $R$-module $M$ in $\C$ or $\D$, there is a natural isomorphism
\[
\Text[\C]{0}{M}{M} \cong \Cext[\D]{0}{M}{M}.
\]
\end{enumerate}
\end{theo}
\begin{proof}
The implication \eqref{theo_19_ii}$\implies$\eqref{theo_19_iii} is clear.

\begin{proof}[(\ref{theo_19_i})$\implies$(\ref{theo_19_ii})] For each $R$-module $C\in\C$, $D\in\D$ and $n\in\ZZ$, one has
\[
\F(N)^{n}(C)=\Cext[\D]{n}{C}{N}=0
\]
by~\cite[Theorem~4.11]{GL22}, and $\G(M)^{n}(D)=\Text[\C]{n}{M}{D}=0$ by~\cite[Theorem~4.9]{GL22}. So it follows from Corollary~\ref{4.5} that $\F(N)^{*}\cong\tF{*}$ and $\G(M)^{*}\cong\tG{*}$. Thus by the isomorphism~\eqref{isomorphism}, one has $\Text[\C]{n}{M}{N} \cong \Cext[\D]{n}{M}{N}$ for all $R$-modules $M$ and $N$, and each $n\in\ZZ$.
\let\qed\relax
\end{proof}

\begin{proof}[(\ref{theo_19_iii})$\implies$(\ref{theo_19_i})]
 We let $C\in\C$ and $D\in\D$. Then one has
\[
\Text[\C]{0}{D}{D} \cong \Cext[\D]{0}{D}{D}=0
\]
by~\cite[Theorem~4.11]{GL22}, and $\Cext[\D]{0}{C}{C}\cong\Text[\C]{0}{C}{C}=0$ by~\cite[Theorem~4.9]{GL22}. Thus $\C$-$\idop_{R}(D)<\infty$ and $\D$-$\pdop_{R}(C)<\infty$ again by~\cite[Theorems~4.9 and~4.11]{GL22}.
\end{proof}
\let\qed\relax
\end{proof}

\section{An application to Tate cohomology}

In this section, we prove a balanced result for Tate cohomology introduced in~\cite{SSW-10a} relying on the complete cohomology discussed before. We set
\[
\Dpdop{(\C)}=\sup\braces[\big]{\Dpdop_{R}{(C)} \st[\big] C \in \C} \qquad \text{and} \qquad \Cidop{(\D)}=\sup\braces[\big]{{\Cidop}_{R}{(D)} \st[\big] D \in \D}.
\]

\begin{lemm}\label{6.1}
Let $(\D,\C)$ be a balanced pair. Then the following statements hold.
\begin{enumerate}\alphenumi
\item \label{ineq_a} If all $R$-modules in $\C$ have finite $\D$-projective dimension, then there is an inequality $\D$-$\pdop(\C) \leq \C$-$\idop(\D)$.
\item \label{ineq_b} If all $R$-modules in $\D$ have finite $\C$-injective dimension, then there is an inequality $\C$-$\idop(\D) \leq \D$-$\pdop(\C)$.
\end{enumerate}
\end{lemm}
\begin{proof}
We only prove~\eqref{ineq_a}; the statement~\eqref{ineq_b} is proved similarly.

Assume that $\C$-${\idop(\D)}=d<\infty$. Let $C$ be in $\C$, and let $\D$-$\pdop_{R}(C) = n <\infty$ by assumption. Thus there exists an $R$-module $M$ such that $\Ext[\DR]{n}{C}{M}\neq 0$ by Lemma~\ref{Chen}. Let $0 \to \Omega^{\D}M \to P \to M \to 0$ be a special $\D$-precover of $M$ with $P\in \D$ and $\Omega^{\D}M\in \D^{\bot}$. Then one gets an exact sequence
\[
\Ext[\DR]{n}{C}{P}\to\Ext[\DR]{n}{C}{M}\to\Ext[\DR]{n+1}{C}{\Omega^{\D}M}
\]
of abelian groups by Sather-Wagstaff, Sharif and White~\cite[Lemma~1.15(a)]{SSW}. Notice that $\Ext[\DR]{n+1}{C}{\Omega^{\D}M} = 0$ by Lemma~\ref{Chen} and $\Ext[\DR]{n}{C}{M}\neq0$, so one gets $\Ext[\DR]{n}{C}{P} \neq 0$. Since $(\D,\C)$ is a balanced pair, one has $\Ext[R\C]{n}{C}{P}\cong\Ext[\DR]{n}{C}{P} \neq 0$. It follows from Lemma~\ref{Chen-dual} that $n\leq d$, as $\C$-$\idop_{R}(P)\leq d$. Thus one has $\D$-$\pdop_{R}(C) \leq d$, which yields $\D$-$\pdop{(\C)} \leq d$.
\end{proof}

\begin{enonce*}[remark]{Tate (co)resolutions}%]\label{resolution and co}
Recall from~\cite{SSW-10a} that a complex $T$ of $R$-modules in $\C$ is \emph{totally $\C$-acyclic} if it is acyclic and the complexes $\Hom{C}{T}$ and $\Hom{T}{C}$ are acyclic for each $R$-module $C\in\C$. Let $N$ be an $R$-module. A \emph{Tate $\C$-coresolution} of $N$ is a diagram $N \qra I \xra{\alpha} T$ wherein $T$ is a totally $\C$-acyclic complex of $R$-modules in $\C$, $N \qra I$ is a proper $\C$-coresolution of $N$, and $\alpha^{n}$ is an isomorphism for $n\gg0$.

Dually, one has the definitions of a \emph{totally $\D$-acyclic} complex $H$ and a \emph{Tate $\D$-resolution} $H \xra{\gamma} P\qra M$ of $M$.
\end{enonce*}

\begin{enonce*}[remark]{Tate cohomology}%\label{TCG}
Let $M$ be an $R$-module admitting a Tate $\D$-resolution $H\longrightarrow P\qra M$. Following~\cite[Definition~4.1]{SSW-10a}, for each $R$-module $N$ and each $n\in\ZZ$, the $n$-th \emph{Tate cohomology} of $M$ with coefficients in $N$ is defined as
\[
\Cext[\DR]{n}{M}{N} = \Hop^n\parens[\big]{\Hom{H}{N}}.
\]

Let $N$ be an $R$-module admitting a Tate $\C$-coresolution $N\qra I\longrightarrow T$. Then for each $R$-module~$M$, the $n$-th \emph{Tate cohomology} of $N$ with coefficients in $M$ is defined as
\[
\Cext[R\C]{n}{M}{N} = \Hop^n\parens[\big]{\Hom{M}{T}}.
\]
\end{enonce*}

\begin{rema}
Assume that $\D$ and $\C$ are closed under direct summands. Then by~\cite[Proposition~4.15]{GL22} the Tate cohomology group $\Cext[\DR]{n}{M}{N}$ is actually the complete cohomology group $\Cext[\D]{n}{M}{N}$ whenever $R$-module $M$ has a Tate $\D$-resolution. Also by~\cite[Proposition~4.14]{GL22}, the Tate cohomology group $\Cext[R\C]{n}{M}{N}$ is actually the complete cohomology group $\Text[\C]{n}{M}{N}$ whenever $R$-module $N$ has a Tate $\C$-coresolution.
\end{rema}

\begin{theo}\label{Tate result}
Let $(\D,\C)$ be a balanced pair. Assume that $\D$ and $\C$ are closed under direct summands and\/ ${\Ext{\geq1}{\D}{\D^{\bot}}}=0={\Ext{\geq1}{{^{\perp}\C}}{\C}}$. Then the following statements are equivalent.
\begin{enumerate}\romanenumi
\item \label{nat_iso_i} All $R$-modules in $\C$ have finite $\D$-projective dimension and all $R$-modules in $\D$ have finite $\C$-injective dimension.
\item \label{nat_iso_ii} $\D$-$\pdop(\C) = \C$-$\idop(\D)<\infty$.
\item \label{nat_iso_iii} Each $R$-module has a Tate $\C$-coresolution.
\item \label{nat_iso_iv} Each $R$-module has a Tate $\D$-resolution.
\end{enumerate}
In this case, for all $R$-modules $M$ and $N$ and each $n\in\ZZ$, there is a natural isomorphism $\Cext[\DR]{n}{M}{N} \cong \Cext[R\C]{n}{M}{N}$.
\end{theo}
\begin{proof}
Since an arbitrary direct sum of $R$-modules in $\D$ is also in $\D$, the invariant $\C$-$\idop(\D)$ is finite if and only if all $R$-modules in $\D$ have finite $\C$-injective dimension. So the statements~\eqref{nat_iso_i} and~\eqref{nat_iso_ii} are equivalent by Lemma~\ref{6.1}.

\begin{proof}[(\ref{nat_iso_ii})$\implies$(\ref{nat_iso_iii})]
Assume that $\D$-$\pdop(\C) = \C$-$\idop(\D)$ = $n <\infty$. Let $M$ be an $R$-module with $M \qra I$ a proper $\C$-coresolution of $M$. Fix a proper $\D$-resolution $P\qra M$ of $M$. Then from~\cite[Proposition~2.2]{Chen} the exact sequence
\[
\dotsb \to P_1 \to P_0 \to M \to 0
\]
is $\Hom{-}{\C}$-exact, as $(\D,\C)$ is a balanced pair. So one gets an exact sequence
\[
\dotsb \to \suspc^{n}P_1 \to \suspc^{n}P_0 \to \suspc^{n}M \to 0
\]
with each $\suspc^{n}P_i\in\C$, which is $\Hom{-}{\C}$-exact. Assembling the above sequence and the $\Hom{-}{\C}$-exact exact sequence
\[
0 \to \suspc^{n}M \to I^{n}\to I^{n+1} \to \dotsb
\]
with each $I^{n}\in \C$, one gets an acyclic complex $T$ of $R$-modules in $\C$
\[
T = \dotsb \to \suspc^{n}P_1 \to \suspc^{n}P_0\to I^{n}\to I^{n+1} \to \dotsb,
\]
which is $\Hom{-}{\C}$-exact. Next we prove that the sequence $T$ is also $\Hom{\C}{-}$-exact. To this end, let $C\in\C$. Then one has $\D$-$\pdop_{R}(C) \leq n$. Let $\Cop^i(T)$ denote the kernel of the homomorphism from $T^{i}$ to $T^{i+1}$ for each $i\in\ZZ$. Note that $T$ is $\Hom{-}{\C}$-exact and each $T_i\in\C$. One gets
\[
\begin{split}
\Ext[R\C]{1}{C}{\Cop^i(T)} &\cong \Extop_{R\C}^{n+1}\parens[\big]{C,\Cop^{i-n}(T)}\\
&\cong \Extop_{\DR}^{n+1}\parens[\big]{C,\Cop^{i-n}(T)}\\
&= 0,
\end{split}
\]
where the first isomorphism holds by~\cite[Lemma~1.15(d)]{SSW}, the second one follows from~\cite[Lemma~2.1]{Chen}, and the equality holds by Lemma~\ref{Chen} as $\D$-$\pdop_{R}(C)\leq n$. Thus the complex $T$ is $\Hom{\C}{-}$-exact, and hence there is a homomorphism $\alpha \colon I \to T$. Then one gets a Tate $\C$-coresolution $M\qra I \xra{\alpha} T$.
\let\qed\relax
\end{proof}

\begin{proof}[(\ref{nat_iso_iii})$\implies$(\ref{nat_iso_i})] For each $R$-module $D\in \D$, there is a Tate $\C$-coresolution $D\qra I \longrightarrow T$ by assumption. $T$ is $\Hom{-}{\C}$-exact, so it is $\Hom{\D}{-}$-exact by~\cite[Proposition~2.2]{Chen}. Thus from~\cite[Proposition~4.14]{GL22}, one has
\[
\Text[\C]{n}{D}{D}\cong \Hop^n\parens[\big]{\Hom{D}{T}}=0,
\]
and so $\C$-$\idop_{R}(D)<\infty$ by~\cite[Theorem~4.9]{GL22}.

Next, for each $R$-module $C\in \C$, we prove $\D$-$\pdop_{R}(C) <\infty$. Let $M$ be an $R$-module. By assumption there is a Tate $\C$-coresolution $M\qra I' \xra{\mu} T'$ of $M$. Then there exists an integer $n>0$ such that $\mu^{j}$ is an isomorphism for each $j\geq n$. Thus for each $i>n$, one has
\[
\begin{split}
\Ext[\DR]{i}{C}{M} &\cong \Ext[R\C]{i}{C}{M}\\
&= \Hop^i\parens[\big]{\Hom{C}{I'}}\\
&\cong \Hop^i\parens[\big]{\Hom{C}{T'}}\\
&= 0,
\end{split}
\]
where the first isomorphism holds as $(\D,\C)$ is a balanced pair. Let
\[
i_{M}=\inf\braces[\big]{n \st[\big] \Ext[\DR]{i}{C}{M} = 0\ \text{for all} \ i>n}.
\]
We claim that there is an integer $s$ such that $i_M\leq s$ for each $R$-module $M$. Actually, it suffices to show that for any sequence $\{M_{t}\}_{t\in\mathbb{N}}$ of $R$-modules there exists such an integer $s$. Set $L = \prod^{\infty}_{t = 0} M_{t}$ and let $s = i_{L}$. Then one has $i_{M_{t}}\leq s$ for each $t\in\mathbb{N}$ as
\[
\braces[\big]{n \st[\big] \Ext[\DR]{i}{C}{L}=0 \ \text{for all} \ i>n} \subseteq \braces[\big]{n \st[\big] \Ext[\DR]{i}{C}{M_{t}} = 0 \ \text{for all} \ i>n}.
\]
Thus one gets that $\Ext[\DR]{s+l}{C}{M} = 0$ for each $R$-module $M$ and all $l>0$, and so $\D$-$\pdop_{R}(C) \leq s <\infty$ by Lemma~\ref{Chen}.
\let\qed\relax
\end{proof}

The implications \eqref{nat_iso_ii}$\implies$\eqref{nat_iso_iv} and \eqref{nat_iso_iv}$\implies$\eqref{nat_iso_i} are proved similarly.

Finally, for all $R$-modules $M$ and $N$, and each $n\in\ZZ$, one has
\[
\begin{split}
\smash{\Cext[\DR]{n}{M}{N}} &\cong \smash{\Cext[\D]{n}{M}{N}}\\
&\cong \smash{\Text[\C]{n}{M}{N}}\\
&\cong \Cext[R\C]{n}{M}{N},
\end{split}
\]
where the first isomorphism follows from~\cite[Proposition~4.15]{GL22}, the second one holds by Theorem~\ref{balance}, and the last one holds by~\cite[Proposition~4.14]{GL22}.
\end{proof}

\section*{Acknowledgments}

We thank the anonymous referee for pertinent suggestions that improved the exposition.

\section*{Declaration of interests}

The authors do not work for, advise, own shares in, or receive funds from any organization that could benefit from this article, and have declared no affiliations other than their research organizations.

\printbibliography

\end{document}